Author's response to reviews

Title: Mediastinitis and sternal prosthesis infection successfully treated by minimally invasive omental flap transposition

Authors:

Valentina Tassi (valevalepg@libero.it)
Silvia Ceccarelli (vicryl83@hotmail.it)
Jacopo Vannucci (jacopovannucci@tiscali.it)
Francesco Puma (fr_puma@yahoo.it)

Version: 2 Date: 29 January 2013

Author's response to reviews: see over
Dear Dr. Zamvar,

we are grateful to the reviewers for the kind appreciation of our manuscript and the fine suggestions to make it better.

We have made any corrections following Prof. Crisci’s statements.

The Discussion has been revised and we added a section to answer the question by Dr. Gudbjartsson about the role of VAC. Moreover we included a new part “Conclusion” as suggested by the Editor where we summarized the most relevant aspects of this report. By following the Reviewers comments we could find a couple of pertinent additional references.

We hope you can find the changes fulfilling the request. Please don’t hesitate to contact me for any additional need or suggestion.

Best regards,

Jacopo Vannucci, on behalf of the Authors